Author’s response to reviews

Title: Early Statin Use in Ischemic Stroke Patients Treated with Recanalization Therapy: Retrospective Observational Study

Authors:

Jioon Kang (rheizer@naver.com)
Nayoung Kim (juli8012@hanmail.net)
Tae Hwan Park (strokezero@gmail.com)
Oh Young Bang (nmboy@unitel.co.kr)
Ji Sung Lee (totoro96a@gmail.com)
Juneyoung Lee (jyleeuf@korea.ac.kr)
Seong-Ho Park (nrpsh@snubh.org)
Moon-Ku Han (mkhan@snu.ac.kr)
Philip B Gorelick (pgorelic@trinity-health.org)
Hee-Joon Bae (braindoc@snu.ac.kr)

Version: 4 Date: 12 May 2015

Author’s response to reviews:

Dear Editor

We highly appreciate the thoughtful comments of the reviewers of this manuscript (BMC Neurology/MS 150063400134493). The authors agree to the points the reviewers have made, have changed the manuscript as requested, and have described all the changes in a point-by-point response to the reviewers’ critiques in the following pages. Also, we have highlighted the changes in the manuscript.

All authors have read and approved the submitted manuscript.

Thank you again for reviewing our manuscript for the journal BMC Neurology.

Sincerely yours,

Hee-Joon Bae, MD, PhD.